THESIS FOR THE DEGREE OF LICENTIATE OF ENGINEERING

Decontamination of Mercury from Aqueous Systems via Electrochemical $PtHg_4$ Alloy Formation

VERA ROTH

Department of Physics CHALMERS UNIVERSITY OF TECHNOLOGY Gothenburg, Sweden, 2023

Decontamination of Mercury from Aqueous Systems via Electrochemical $\rm PtHg_4$ Alloy Formation

Vera Roth

© Vera Roth, 2023

Division of Chemical Physics Department of Physics Chalmers University of Technology SE-412 96 Göteborg, Sweden Phone: +46(0)31 772 1000

Cover: Electrochemical alloy formation between mercury ions (yellow) and platinum (grey), forming $PtHg_4$ (grey and blue). Image by Rasmus Svensson.

Printed by Chalmers Digitaltryck, Gothenburg, Sweden 2023.

i

Decontamination of Mercury from Aqueous Systems via Electrochemical $PtHg_4$ Alloy Formation

VERA ROTH Department of Physics Chalmers University of Technology

Abstract

Mercury is worldwide a severe heavy metal pollutant and a serious threat to human health and ecosystems. Mercury can be released into the environment by various natural processes; however, mercury pollution has been significantly amplified by human activities such as mining, production of fertilizers, waste incineration and disposal. Once mercury enters aquatic ecosystems, it bioaccumulates and biomagnifies, and mercury can travel vast distances in water, resulting in widespread contamination and devastation. It is thus crucial for the well-being of humans and ecosystems to address the mercury contamination of water. It is important not only to reduce the mercury pollution at its source, but also reduce the present contamination of water. Current technologies used to mitigate the situation are far from optimal and the development of more efficient and effective removal methods are needed. A novel method based on electrochemical alloy formation between mercury ions in solution and a platinum surface has been shown to offer various advantages over previous techniques. The method can be used to decontaminate both waste and natural waters, as well as highly corrosive acids generated from mining industries.

In this thesis, the fundamental principles of electrochemical $PtHg_4$ alloy formation for mercury removal are studied. The reaction mechanisms and energetics are examined, and its potential for future large-scale applications in industries is explored. It is shown that mercury can be efficiently removed from industrial concentrated sulfuric acid derived from zinc smelting. Comparison between laboratory-scale and a medium-scale demonstrator show excellent scalability of the removal method. Additionally, mercury is efficiently removed from diluted acids with initial mercury concentrations as low as 0.25 µg/L, making the method applicable for both environmental and industrial settings. This thesis establishes an important foundation for further research and optimization of electrochemical alloy formation as a mercury removal method. With continued development and advancements, the method has the potential to contribute significantly to the mitigation of global mercury pollution for the protection of human health and ecosystems.

Keywords

Mercury Removal; Electrochemistry; Platinum; Mercury; Alloy Formation

List of Publications

This thesis is based on the following publications:

Paper I

Roth, V., Järlebark, J., Ahrnens, A., Nyberg, J., Salminen, J., Vollmer, T. R., and Wickman, B.

"Mercury Removal from Concentrated Sulfuric Acid by Electrochemical Alloy Formation on Platinum"

ACS ES&T Engineering 3, 823 (2023).

Paper II

Feldt, E., Järlebark, J., **Roth, V.**, Svensson, R., Gustafsson, P. K., Molander, N., Tunsu, C., and Wickman, B. "Temperature and concentration dependence of the electrochemical PtHg₄ alloy formation for mercury decontamination"

Separation and Purification Technology **319**, 124033 (2023).

Paper III

Roth, V., Valter-Lithander, M., Järlebark, J., Bilesan M. R., and Wickman, B.

"On the Mechanism of Electrochemical Alloy Formation Between Mercury and Platinum for Mercury Removal from Aqueous Systems" In Manuscript

Author Contributions

Paper I

I contributed to the design of the experimental process and preparation of the working electrodes. I performed ICP-MS analyses, prepared the first draft of the manuscript, and was involved in the finalization of the manuscript.

Paper II

I handled the finalization of the manuscript, which included the final version of the figures and presentation of the data.

Paper III

I contributed to the design of the experiments and was responsible for the preparation of the working electrodes. I conducted EQCM measurements and ICP-MS analyses, and prepared the first draft of the manuscript.

Contents

Ab	stract	iii					
Lis	of Publications	\mathbf{v}					
Aι	Author Contributions v						
1	Introduction	$\frac{1}{2}$					
2	Mercury2.1Sources2.2Environmental Impact2.3Removal Methods	3 3 5 6					
3	Electrochemistry3.1Fundamentals of the Electrochemical Cell3.2Thermodynamics3.3Kinetics	11 11 12 13					
4	Methods 4.1 Electrochemical Alloy Formation 4.2 Electrochemical Measurements 4.2.1 Electrochemical Quartz Crystal Microbalance 4.2.2 Batch Experiments 4.3 Analytical Techniques 4.3.1 Inductively Coupled Plasma Mass Spectrometry	15 15 17 17 18 20 20					
5	Results 5.1 Reaction Mechanism 5.1.1 Alloy Formation 5.1.2 Regeneration 5.1.3 Temperature and Concentration Studies 5.2 Large-scale Mercury Decontamination	 23 24 26 28 30 					
6	Conclusion and Outlook	35					

Acknowledgment	
----------------	--

39

Chapter 1 Introduction

Environmental pollution is one of the most urgent and rapidly escalating issues facing the modern world. Across the globe, toxic chemicals are incessantly released into the environment, leading to detrimental impacts on soil, air and most critically, water. Water is essential to all forms of life, and the implications of its pollution are far-reaching and devastating. Contaminated water impacts entire ecosystems, and becomes in many cases a vector for disease and death [1-3].

Mercury, which is a highly toxic heavy metal, has been identified as one of the current top six most toxic threats in the world. According to the Green Cross, mercury is surpassed only by lead and radionuclides in terms of the number of people affected, the severity of exposure, and the geographical extent of its contamination [4]. In addition, the World Health Organization (WHO) has listed mercury among the top ten chemicals of major public health concern [5]. The significant impact of mercury is further highlighted by the fact that the element is the sole focus of a global environmental treaty, the Minamata Convention on Mercury, which emphasizes the severity of its pollution and the urgency of global recognition [6].

Mercury is found naturally in the environment and is emitted into the atmosphere through natural processes such as rock weathering, geothermal activity and volcanic eruptions [7]. However, mercury emissions from anthropogenic sources exceed the natural ones by more than one order of a magnitude, with over 2500 tonnes being emitted annually through activities such as mining, fertilizer production, waste incineration and disposal. Through human activities alone it is estimated that a total of 1,540,000 tonnes of mercury was released between the years 1850 - 2010, with 470,000 tonnes directly emitted into the atmosphere and 1,070,000 tonnes released into land and water systems [8]. Once mercury is emitted into the atmosphere, it can travel vast distances before being deposited onto land and water bodies. Within these environments, mercury can either be sequestered within the soils or sediments, or be re-emitted back into the atmosphere. Mercury's unique cycling ability, allowing it to continuously be re-emitted from land and water surfaces, means that mercury released decades ago is still circulating in today's environment [9]. Mercury can persist in the environment for millennia, remaining mobile and posing a toxic threat until its ultimately bound in deep ocean sediments or subsurface soils [10]. Water plays a key role in the mobility and distribution of mercury, contributing to approximately 60% of its environmental cycling [8]. When mercury enters water bodies, it can bioaccumlates in aquatic organisms and subsequently biomagnify up the food chain, posing a significant threat not only to humans but also to all aquatic and terrestrial wildlife [11]. Therefore, addressing the issue of mercury pollution in water is of paramount importance for both global environmental sustainability and public health. There is an urgent need to develop effective strategies to manage and mitigate the impact of mercury.

Current established methods for mercury removal from aqueous solutions, although functional, are far from optimal. The main methods include techniques such as adsorption, filtration, ion exchange, and precipitation, which, depending on the specific context and application, have their own advantages and disadvantages [12, 13]. Common limitations include low selectivity for mercury, high capital cost and energy consumption, technical complexity and the generation of secondary waste. Moreover, the effectiveness of these methods can be greatly influenced by the pH of the solution, and to date, no method has proven effective for mercury removal from concentrated acids [13, 14].

Recently, a promising new method for mercury removal from aqueous solutions has been presented, based on electrochemical alloy formation between mercury ions in solution and a platinum surface [15]. This technique holds the potential to overcome the limitations of existing removal techniques and provide a more effective and efficient approach to mercury removal [15, 16]. With further research and optimization, this removal method could play a significantly role in mitigating the environmental and health impacts associated with mercury contamination in water systems.

1.1 Scope of the Thesis

This thesis presents research and findings aiming to further enhance our understanding of electrochemical alloy formation as a method for mercury removal from aqueous systems. Through electrochemical experiments and detailed analysis, the reaction mechanisms and energetics of the electrochemical process are studied. Furthermore, the potential of this method for large-scale mercury removal is explored. Chapter 2 provides a background on mercury pollution, detailing its sources, impacts, and various removal methods. Chapter 3 presents the fundamentals of electrochemistry and the structure of an electrochemical cell, along with the underlying thermodynamics and kinetics of the electrochemical processes. Chapter 4 outlines the experimental methods and analytical techniques used in this study, while Chapter 5 discusses and expands upon some of the results from **Paper I** through **III**. Lastly, Chapter 6 concludes the thesis and gives suggestions for future research directions.

Chapter 2

Mercury

This chapter provides an overview of mercury pollution, examining its natural and anthropogenic sources, as well as its impact on both human health and ecosystems. In addition, it presents the current methods used for mercury removal from aqueous systems.

2.1 Sources

Mercury (Hg) is a naturally occurring element and is widely distributed in the earth's crust, oceans and atmosphere [17]. Despite its natural occurrence, human activities have significantly increased its abundance, leading to elevated levels of mercury pollution in the environment [18]. In recent decades, efforts have been made to reduce mercury emissions. Yet, the persistence of mercury in the environment remains a significant issue, as the mercury released years ago is still actively circulating in our environment. The issue of mercury emissions are of particular concern due to its high toxicity and the rapid pace at which it spreads and accumulates in natural water bodies [19].

Sources of mercury pollution can be broadly classified into two main categories [8]: natural and anthropogenic, which can be further subdivided into primary and secondary sources. Primary natural sources, such as volcanoes, rock weathering and geothermal activities, are estimated to release approximately 80 - 600 tonnes of mercury annually. In stark contrast, primary anthropogenic sources, originating from human activities, emit around 1900 - 2900 tonnes of mercury into the environment annually. It is further estimated that a total of 1,540,000 tonnes of mercury has been released by human activities prior to 2010, with 70% of these emissions occurring after 1850 [8]. This notable increase in mercury emissions can be linked to the industrial revolution and its rapid expansion of industrial processes such as mining, burning of fossil fuels, and waste incineration [20]. Today, the main activities increasing the total pool of mercury in the environment are processes such as burning of fossil fuels (49%), artisanal and small-scale gold mining (ASGM) (14%), mining of non-ferrous metals (11%), cement production (8%) and waste disposal (6%) [7,

21].

Once mercury is released into the environment, it can travel vast distances via air and water currents [11]. Mercury emitted in one region of the world can reach remote areas, such as the Arctic [22, 23]. After deposition, mercury cycles through a variety of environmental surfaces such as soil, vegetation and water bodies, which continuously re-emit mercury back into the atmosphere. This re-emission of mercury is often categorized as a secondary emission source and can originate from both primary anthropogenic and natural sources [10, 24]. Essentially, primary emissions increase the overall mercury pool in the environment, while secondary emissions redistribute it within ecosystems, resulting in widespread pollution of both terrestrial and aquatic environments. Mercury can persist in the environment for thousand of years, remaining mobile and cycle between re-emission and deposition, until its eventually bound in its final forms in deep ocean sediments or subsurface soils [25]. Estimates show that secondary emissions can be up to three times greater in magnitude than those from primary anthropogenic sources, with water playing a key role, contributing to more than 60% to the environmental cycling of mercury [18, 26]. Collectively, it is estimated that approximately 8400 tonnes of mercury is released annually into the atmosphere from all sources, with 8000 tonnes deposited on land and water bodies. As a results, approximately 950,000 tonnes of mercury is accumulated in soils and 310,000 tonnes in oceans [27]. A more detailed breakdown of different mercury sources is depicted in Fig. 2.1.



Figure 2.1: The annual cycle of estimated mercury emissions (in tonnes, for the year 2015), deposition, and re-emission (red arrows) in the environment. In addition, the total mercury estimated to be accumulated in various reservoirs is shown, including organic and mineral soils, as well as surface ocean, intermediate water, and deep waters. The data is obtained from Ref. 27.

2.2 Environmental Impact

Mercury is an element with physical and chemical properties that contribute to its complex and extensive effects on all forms of life and their ecosystems. Its high toxicity, mobility, and ability to bioaccumulate and biomagnify through the food chain make it a particularly concerning environmental pollutant [28]. In particular, aquatic ecosystems demonstrate extreme vulnerability to the impacts of mercury pollution. Mercury, once it enters bodies of water, can travel up the food chain, becoming progressively concentrated in fish, birds and mammals [18]. As a result, even relatively low levels of mercury in water can evolve into a more menacing threat, e.g., the concentration of mercury in tissues of top predators such as whales and seals, can exceed million times the levels in their surrounding waters [10]. The species affected can develop behavioral changes, reproductive issues, and in severe cases their populations may decline or even be driven towards extinction [29, 30].

The adverse effects of mercury pollution not only impact wildlife but also extend to the human population, see Fig. 2.2. Humans are primarily exposed to mercury through the consumption of contaminated fish and seafood, as well as occupational exposure in industries [31]. Short term exposure to high levels of mercury, mainly through inhalation in occupational settings, can result in mercury poisoning. Initial symptoms are often flu-like and can include soar throat, coughing, chest pain, nausea, headaches and vision problems. In severe cases, mercury poisoning can lead to respiratory and kidney failures, and may even prove fatal. Long term effects of mercury exposure typically stem from chronic exposure to lower levels of mercury largely through consumption of contaminated food, and can cause neurological and cardiovascular damage, potentially leading to life threatening conditions [10, 31, 32]. The most alarming aspect of long term mercury exposure may be its impact on developing children and fetuses. Their nervous system is particularly vulnerable to mercury toxicity, potentially resulting in cognitive impairments and learning disabilities [33–35].

In response to the extensive toxic effects of mercury, various global policies and initiatives have been set to raise public awareness and mitigate mercury pollution on a global scale. In 2015, the environmental organization Pure Earth together with Green Cross defined mercury as one of world's most toxic threats, alongside lead, radionuclides, chromium, pesticides and cadmium [36]. It was reported that approximately 20 million people worldwide are at risk from mercury exposure, with an estimated 1.5 million years of healthy life (Disability Adjusted Life Years, DALYs) lost due to its effects. Furthermore, the World Health Organization (WHO) has recognized mercury as one of the top ten chemicals of public health concern, listed along with chemicals such as arsenic, lead asbestos and hazardous pesticides [5]. As a results, both WHO and the European Union (EU) have implemented regulations for mercury concentration in drinking water, where only water with mercury concentration below 6 µg/L and $1 \mu g/L$, respectively, is considered safe for consumption [5, 37]. In 2017, the Minatamata Convention on mercury came into force, which strives to reduce the harmful effects of mercury. This global treaty involves coordinated efforts



Figure 2.2: Short term and long term health effects of human exposure to mercury. The data is obtained from Ref. 10 and Ref. 31.

to minimize the supply and trade of mercury, reducing its use, and manage emissions [6].

2.3 Removal Methods

Various strategies have been employed to mitigate mercury pollution, such as mercury capture technologies, phasing out the use of unnecessary mercurycontaining products and robust waste management systems. Despite these efforts, mercury continues to be a pervasive environmental issue [38].

Existing methods for mercury decontamination of aqueous systems are far from ideal and often prove ineffective under certain conditions, particularly in highly acidic and corrosive solutions [12, 13]. Traditional removal methods include adsorption, coagulation and flocculation, flotation, ion exchange resins, membrane filtration and chemical precipitation. Each of these techniques have their own set of advantages and limitations, as detailed in Table 2.1. Notably, the main limitations include high energy demands and operational costs, low selectivity, and a strong dependency on pH levels [39, 40]. Currently, no commercial method has the ability to operate in concentrated acids, e.g., for effective sulfide precipitation a pH range of 7-9 is required, and the optimum pH for adsorption methods is typically between 4 - 5 [41]. Furthermore, most materials used for ion-exchange, membrane separation, and adsorption methods, degrade when used to treat concentrated acids [13]. There is thus an urgent need for the development and implementation of new and improved solutions for mercury removal from various aqueous systems. Electrochemical alloy formation, though still in the early stages of development, presents a promising alternative to established methods [15]. A comparison of its primary advantages and drawbacks, relative to the other methods, is provided in Table

2.1. The fundamental principles, advantages, and limitations of the method are discussed in more detail in the following chapters of this thesis.

Adsorption is a widely used technique for mercury capture, particularly effective in treating vapor and wastewater streams However, the mercury removal capacity can greatly vary among different types of adsorbent. While activated carbon is the most commonly used adsorbent, cost-effective alternatives such as agricultural waste-based adsorbents and biosorbents, including microbes, algae, and non-living biomass, have been proposed. The efficiency of this removal technique is influenced by factors such as pH, temperature, and the saturation level of the adsorbent. Furthermore, the method shows limited selectivity for mercury, as other pollutants can also be adsorbed. The regeneration of the adsorbent material, although not always possible, can be expensive and may result in the degradation of the material [40, 42].

Coagulation and flocculation, commonly employed to treat wastewater, involve the addition of a coagulant to react with mercury ions, creating larger insoluble particles. Despite the methods simplicity and cost-effectiveness, the resulting particles need to be removed via additional methods, such as precipitation or filtration [38, 40].

Flotation is another technique used to extract pollutants from water, with ion flotation being particularly effective for the removal of heavy metal ions, such as mercury. The process involves adding an amphiphilic surfactant to the water, where the surfactant with the mercury ion will attach to air bubbles. The method's selectivity and efficiency can be high, although both these factors are pH dependent. In addition, the costs of installation and maintenance can be high [38, 43].

Ion exchange treatment, a process in which mercury ions are swapped with other non-toxic cations within an ion exchange resin, provides a quick and efficient approach to mercury removal. This method is particularly valuable for treating larger volumes of solutions containing low concentrations of mercury ions. However, the efficiency of this process is heavily influenced by factors such as pH, temperature, and contact time, and due to rapid resin saturation the technique is less applicable for higher concentrations of mercury [40, 44].

Membrane filtration includes a variety of technologies that utilize different types of membranes for wastewater treatment. One such technology is ultrafiltration, where low pressure is applied to force water through a membrane, separating molecules and colloidal particles larger than the membrane's pore size. In this process, ligands are introduced to the water to form larger chelates with mercury. However, due to the substantial cost associated with these ligands, their recovery and reuse are crucial to make this technique economically feasible. Other membrane filtration techniques are also reported to have rapid clogging of the membranes and removal efficiencies affected by pH [38]. Precipitation is the most frequently used method for industrial wastewater treatment. The process involves the addition of a chemical, typically a sulfide salt, to water, leading to the formation of insoluble precipitates like mercury sulfide. The effectiveness of this process is heavily dependent on pH, as well as not being selective for mercury, as it may cause other metals to precipitate as well [13, 38].

Table 2.1: Methods for removing mercury from aqueous solutions, along with their main advantages and limitations [13–15].

Method	Advantages	Limitations	
Adsorption	Simple operation Low secondary waste High removal rate	High capital cost Low selectivity Secondary waste pH dependant	
Coagulation/ Flocculation	Simple process Low capital cost High removal efficiency	Chemical consumption Secondary waste Ineffective at low/high conc. Secondary waste pH dependant	
Flotation	High selectivity High removal efficiency	High capital cost High energy consumption pH dependant	
Ion exchange	Simple operation Time efficient High removal efficiency	High capital cost Low selectivity Secondary waste pH dependant	
Membrane filtration	Simple process Time efficient	High operation cost High energy consumption pH dependant Membrane clogging	
Precipitation	Simple operation Cost efficient High removal rate	Low selectivity Ineffective at low conc. Secondary waste pH dependant	
Electrochemical alloy formation	Simple operation low energy consumption High removal efficiency High selectivity Independent of pH	Early development stages	

Chapter 3 Electrochemistry

In this chapter, the fundamentals of electrochemistry are presented, along with the structure of the electrochemical cell. The underlying thermodynamics and kinetics of electrochemical processes are also discussed.

3.1 Fundamentals of the Electrochemical Cell

Electrochemistry is a rapidly evolving branch of science that studies the relationship between electricity and chemical reactions. Electrochemistry has proven to be instrumental in many technological advancements, with applications in battery technology, fuel cells, electroplating, and corrosion processes [45]. Furthermore, electrochemistry presents new possibilities in environmental remediation, removing toxic heavy metals such as mercury from various aqueous systems [15, 16].

At the core of electrochemistry are oxidation-reduction (redox) reactions, in which electrons are transferred from one chemical species, the reducing agent, to another, the oxidizing agent. These reactions can be either spontaneous, generating electrical energy as in galvanic cells, or non-spontaneous, where electrical energy is used to drive a desired chemical reaction, as in electrolytic cells [45]. An electrochemical cell consists of two electrodes, the anode and the cathode, which are connected by an internal ionic pathway provided by an electrolyte and an external electric connection. Each electrode serves as the location of one of two half-reactions that occur at the interface between the electrode and the electrolyte. Electrons flow from the anode, where the oxidation occurs, to the cathode, where reduction takes place. For simplification, two electrodes within an electrochemical cell can be referred to as the working electrode (WE) and the counter electrode (CE). The WE is the site where the half reaction of interest takes place, and the CE serves as either the anode or cathode to counterbalance the reactions at the WE. A potential or current is applied between these two electrodes, and the resulting response is measured. However, to measure absolute potentials, a third electrode, a reference electrode (RE) can be introduced to the system. As no current passes through the RE, it

maintains a constant potential, and the potential is then measured relative to the RE. The current, meanwhile, is between the WE and the CE, as illustrated in Fig. 3.1 [45].



Figure 3.1: (a) A typical 2-electrode cell and (b) 3-electrode cell setup, including a working electrode (WE), counter (CE) and a reference electrode (RE).

3.2 Thermodynamics

A simple one step redox reaction, can be expressed with the following reaction, where an oxidizer O_x in a solution is reduced to R_{ed} by accepting n-electrons (e⁻) [45]:

$$O_x + ne^- \rightleftharpoons R_{ed}$$
 (3.1)

The reaction quotient (Q_r) represents the ratio of the activities (a) of the reduced species to the oxidised species. When these activities are near unity, the Q_r can be approximated by the ratio of their respective molar concentrations ([]):

$$Q_r = \frac{a_{red}}{a_{ox}} = \frac{[red]}{[ox]} \tag{3.2}$$

The Gibbs free energy (ΔG) can be defined as the maximum amount of work (-W_{max}) that can be extracted from a system, and can be written as a function of the cell potential (E_{cell}) and the total charge transferred (nF), where F is the Faraday's constant:

$$\Delta G = -W_{max} = -nFE_{cell} \tag{3.3}$$

The redox reaction is spontaneous if $\Delta G < 0$, and non-spontaneous if $\Delta G > 0$. ΔG can be further related to the Gibbs free energy under standard conditions (ΔG^0) and Q_r , where R is the gas constant and T the absolute temperature:

$$\Delta G = \Delta G^0 + RT lnQ_r \tag{3.4}$$

When $\Delta G = 0$ and the redox reaction is at equilibrium, Eq. 3.3 and 3.4, can be rearranged to yield the Nernst equation for the reaction, where E_{cell}^0 is the cell potential under standard conditions:

$$E_{cell} = E_{cell}^0 - \left(\frac{RT}{nF}\right) lnQ_r \tag{3.5}$$

While the Nernst equation provides crucial information about equilibrium states, it does not provide information on how quickly a system reaches equilibrium, as reactions may not reach equilibrium due to kinetic limitations. Therefore, studying the reaction kinetics, particularly at the electrode surface, is crucial for a deeper understanding of electrochemical reactions [45].

3.3 Kinetics

The rate (r) of an electrochemical reaction can be defined as a function of the rate constant (k), the concentration of the reactant (C) and the reaction order (α) [45]:

$$r = kC^{\alpha} \tag{3.6}$$

The rate constant of a reaction can be greatly influenced by temperature, as described by the Arrhenius equation [45]:

$$k = A e^{\frac{-E_A}{k_B T}} \tag{3.7}$$

In this equation, T represents the absolute temperature, A is the pre-exponential factor, k_B stands for the Boltzmann constant, and E_A is the apparent activation energy of the reaction. The apparent activation energy can be determined by measuring the reaction rate at various temperatures and at a constant concentration of the reactant.

Chapter 4 Methods

This chapter presents the electrochemical methods and analytical techniques that form the foundation for the studies detailed in **Papers I**, **II**, and **III**.

4.1 Electrochemical Alloy Formation

When elemental mercury (Hg^0) comes in contact with a noble metal such as platinum (Pt), a stable metallic alloy can form. This phenomenon has been extensively studied in the past [46–51]. However, its potential as a technique specifically for mercury removal from aqueous solutions has only recently been presented [15, 16].

To remove mercury, predominantly found in ionic form in contaminated waters, a sufficiently reductive potential is applied to a platinum electrode immersed in a mercury-ion containing solution, inducing the formation of an alloy in the thermodynamically favored phase, PtHg₄. The overall reaction of this formation can be written as follows:

$$Pt + 4Hg^{2+} + 8e^- \to PtHg_4 \tag{4.1}$$

The alloy formation is believed to be a multi step process, in which intermediates such as PtHg and PtHg₂ can form [15]. The process can be described in three main steps, as illustrated in Fig. 4.1: (i) Reduction of the mercury ions in solution on the platinum electrode surface. (ii) The insertion of elemental mercury into the Pt lattice to form the PtHg₄ alloy. (iii) Elemental mercury atom diffuses through the PtHg₄ to grow the alloy.

This new removal method has the potential to offer several significant advantages over other current available methods, as detailed in Section 2.3. It has previously been demonstrated that the method exhibits high selectivity for mercury and is efficient at both high and low initial concentrations of mercury, remaining unaffected by pH in the range of 0 - 6.6. In addition, the process consumes small amounts of energy and is reversible, allowing regeneration of the platinum electrodes and ensuring safe recovery of mercury [15]. Once the electrodes are saturated with mercury, they can be transferred and regenerated by applying a positive potential in a small volume of solution specifically designed for regeneration. When the concentration of mercury in the regeneration solution reaches a high level, established removal methods such as precipitation can be utilized to recover mercury with high purity. In theory, other metals could serve as alternatives to platinum for the electrode material. Copper (Cu), being more abundant and cost-effective than platinum, could be an option; however, its instability in oxidizing acids and water without potential control, coupled with its inability to be regenerated for reuse, limit its applicability [16]. Gold (Au), another potential alternative, can form a stable amalgam with mercury, but its higher cost and lower mercury capacity in its most stable alloy compared to platinum (Au₃Hg vs. PtHg₄) render it a less efficient choice [52]. The usage of less noble metals such as zinc (Zn), aluminium (Al), and tin (Tn) is also an option, however their proneness to oxidation and degradation hinders efficient alloy formation, making them unsuitable in practice [53, 54].



Figure 4.1: The alloy formation process, where a divalent mercury ion is reduced on the electrode surface and inserted into the alloy $PtHg_4$, followed by a diffusion through the alloy and a reaction with platinum at the alloy-platinum interface.

A more in-depth examination of the method is needed to further elucidate the exact reaction mechanisms of the alloy formation, to understand the precise thermodynamics and kinetics of the process, as well as the intricacies of the regeneration process. **Papers I**, **II**, and **III** delve into various aspects of the removal method, including the formation, growth, and dissolution of the alloy, offering crucial insights for the optimization of both the alloying and regeneration processes. Therefore, these investigations represent a vital step

towards establishing this technique as a viable solution for mercury removal from aqueous solutions.

4.2 Electrochemical Measurements

Electrochemical experiments, focusing on the alloy formation between a platinum electrode and mercury in solutions, are carried out to examine the mercury removal process. Various experimental techniques are used to examine different aspects of the method, from its fundamental principles to its more applied features.

4.2.1 Electrochemical Quartz Crystal Microbalance

Electrochemical quartz crystal microbalance (EQCM) is a technique that combines the principles of electrochemistry and piezoelectricity to investigate in real-time various electrochemical processes [55, 56]. In particular, EQCM can precisely measure mass changes in thin films on an electrode during processes such as electrodeposition and dissolution of metal films. By using EQCM, kinetic examination of processes that involve events at the monolayer level and thin films are able to be examined, which would otherwise remain undetected by conventional electrochemical methods [57].

The basis of EQCM operation lies in a quartz crystal sensor, coated on both sides with electrodes. When an alternating electric field is applied, the crystal oscillates at a specific resonant frequency (Δf_0). The frequency of oscillation is highly sensitive to the mass of the crystal; the frequency will shift if a mass is added or lost [55]. The Sauerbrey equation (4.2) can be used to relate the frequency shifts of the crystal to the mass change, where f_0 is the resonant frequency of the the sensor before any mass change, n is the overtone, A_s is the active surface area of the electrode, μ_q is the shear elastic modulus and ρ_q is the sensors density of the quartz [58].

$$\Delta f_n = -\frac{2f_{0,n}^2 n}{A_s \sqrt{\rho_q \mu_q}} \Delta m \tag{4.2}$$

For the Sauerbrey equation to be valid, certain conditions need to be met and any deviations from these conditions can lead to inaccuracies in the calculated mass changes. The deposited or removed film must be considered rigid, contribute to less than 2% of the mass of the crystal sensor, and be uniformly distributed on its surface. A film is considered rigid if the frequencies shift of different overtones overlap, with minimal energy losses and dissipation changes (ΔD) close to zero. A high ΔD indicate that the surface is soft or viscous, and thus viscoelastic models should be used instead of the Sauerbrey equation. The ΔD can be written according to Eq. 4.3, where Q_s is the sum of all energy losses in the system per oscillation cycle. The Q_s factor can be further defined by f₀ over the resonance peak bandwidth BW [55].

$$\Delta D = \frac{1}{Q_s} = \frac{BW}{f_0} \tag{4.3}$$

In **Papers II** and **III**, EQCM is used to monitor in real time, both the alloy formation and the dissolution of the alloy. A typical EQCM setup used for electrochemical measurements can be seen in Fig. 4.2. The WE is a planar AT-cut quartz crystal sensor mounted in a dip holder, exposing only the top platinum electrode to the electrolyte solution.



Figure 4.2: A typical 3-electrode setup for EQCM measurements. A quartz crystal sensor, mounted inside a dip holder, is used as the WE, with a platinum wire CE, and a glass RE.

4.2.2 Batch Experiments

Batch experiments are used to study the mercury removal process at laboratory scale or for scaled-up experiments more relevant for industrial applications. These experiments are distinct from in situ experiments, such as those conducted using EQCM, as they do not involve real-time measurements. During batch experiments, the concentration of mercury in the electrolyte solution, before, during and after electrochemical treatment, is directly measured using an analytical technique, namely Inductively coupled plasma mass spectrometry (ICP-MS).

Batch experimental setups can involve a 2-electrode or a 3-electrode configuration, and the selection of the WE is dependent on the study's objectives. When the focus is on understanding the fundamentals of an electrochemical process, such as reaction mechanisms, simple designs like thin silica glass plates coated with platinum are often used at lab-scale (Fig. 4.3). In contrast, for studies that require a WE with a larger surface area or resilience under harsh conditions, steel or carbon foam-supported platinum WEs can be employed [59, 60]. Foam WEs can be particularly advantageous for large-scale batch experiments in concentrated acids, where fast and efficient mercury removal is the primary objective. A typical batch experimental setup, at lab-scale (50 mL) and a scaled-up version (20 L), is presented in Fig. 4.4.



Figure 4.3: A silica glass WE, and a steel/carbon foam WE.



Figure 4.4: A typical 3-electrode setup for batch experiments at lab-scale (50 mL), consisting of a foam WE, diamond CE, and a glass RE. In addition, a scaled-up version (20 L) using a 2-electrode setup, with two foam WE and two diamond CE.

In **Papers II** and **III**, batch experiments at lab-scale are conducted using a glass WE to study the reaction mechanism of the removal method. Where in **Paper I**, batch experiments at lab-scale and a scaled-up version are used

to evaluate the removal efficiency and stability of different foam WEs, to investigate the removal method's potential for large-scale applications.

4.3 Analytical Techniques

4.3.1 Inductively Coupled Plasma Mass Spectrometry

Inductively coupled plasma mass spectrometry (ICP-MS) is a commonly used technique for measuring trace and ultra-trace levels of elements in a variety of matrices, including aqueous solutions. ICP-MS has diverse applications, prominently in fields such as environmental and pharmaceutical monitoring, clinical research, and toxicological studies Recently there has been a significant shift towards the use of ICP-MS in clinical laboratories, owing to the many advantages it can offer over other common techniques, such as atomic absorption (AAS) and atomic emission spectroscopy (AES). The main advantages of ICP-MS include multi-element detection, low detection limit and sample volume requirements. However, the initial cost of ICP-MS instrument can be high, as well as the operating costs themselves, as the instruments requires argon gas and other high purity gases. Nonetheless, the advantages of the ICP-MS often outweigh these expenses, particularly in settings where precise and accurate trace element analysis is essential [61].

The operation of an ICP-MS instrument can be described as follows: The sample, typically an acidic solution, is introduced into the system via a peristaltic pump and is nebulized into a fine aerosol. This aerosol is transported into a plasma torch, where the sample is atomized and subsequently ionized. The ions generated in the plasma are extracted and guided into a mass analyzer through a set of electrostatic lenses or ion optics. In the mass analyzer, the ions are separated based on their mass-charge (m/z) ratios and the abundance of different isotopes are measured in the detector, which allows for the quantification of the elements present in the original sample. A simplified illustration of the instrument can be seen in Fig. 4.5.



Figure 4.5: Schematic illustration of an ICP-MS instrument, based on data from Ref. 61.

Mercury can be very difficult to analyse in solutions. This extremely volatile element is notorious for its so-called memory effect, where a prolonged signal count for mercury is sustained long after the initial sample has been analyzed and an appropriate washout time carried out [62–64]. This effect can lead to a non-linear calibration of the instrument, reduced sensitivity over time, and a high dependency of the signal on the sample matrix. Even at low mercury concentrations the effect can be prevalent, as mercury can adhere to the walls of both the ICP-MS transfer tube and the spray chamber. Divalent mercury is a soft acid metal cation, and its covalent, and occasionally nonpolar behaviour, allows it to adhere strongly to nonpolar surfaces such as the polymeric tubing of the sample introduction system. In addition, divalent mercury is easily reduced, and in its elemental form, it is neither lipophilic or hydrophilic, and it can easily stick to any crevices, making it difficult to rinse out [65].

Several steps have been proposed to eliminate the issues associated with the memory effect of mercury, such as the addition of gold and stabilizing agents to the sample, as well as specific acid rinsing solutions [65, 66]. Particularly, for stabilization against mercury reduction during analysis, hydrochloric acid (HCl) can be used to prepare the sample matrix. By using HCl, HgCl₂ present in the sample can be adsorbed on hydrophobic surfaces and if sufficient concentrations of HCl is present, mercury can be found predominantly in the water soluble anionic form $[HgCl_4]^{2-}$. If the concentration of the chloride ions is too high, the signal count can however be suppressed by Cl⁺ and ArCl⁺ species formed in the plasma [65].

In **Papers I** and **II**, ICP-MS is used to quantify mercury and other metals of interest in solution. The samples taken from the electrochemical measurements are diluted using 1 M HCl to fall within the ICP-MS instrument's detection limits, ensuring the stability of mercury in the sample and guarantee signal accuracy. Moreover, the standard matrices for the calibration curve are carefully prepared to exactly mirror the samples taken from the electrochemical measurements. Internal standards can be added to all samples to further correct for random errors, such as noise, torch instabilities, and matrix effects. These internal standards possess atomic mass and ionization potential similar to the element of interest, allowing any deviations to be accounted for [67]. By taking these measures, the memory effect during mercury analysis with ICP-MS can be minimized, leading to negligible deviations and eliminating the need for further preventative measures to guarantee precise and reliable results.

Chapter 5

Results

In this chapter, some of the results from **Papers I**, **II**, and **III** are presented.

5.1 Reaction Mechanism

To better understand and optimize electrochemical alloy formation as mercury removal method, it is crucial to gain insights into various aspects of its reaction mechanism. These include, but are not limited to, the onset potential of the alloy formation and its dissolution, as well as kinetic parameters such as its reaction order in mercury ion concentration and the activation energy for the alloy formation.

It is important to note, that there is currently no universally accepted or standardized method for determining the onset potential of an electrochemical process [68]. One commonly used approach is the tangent method, where a tangent line or a linear extrapolation of the current at higher overpotentials is drawn, and the potential at which this extrapolation intersects the baseline is defined as the onset potential. Another approach is the de-visu method, which relies more on subjective visual assessment than the tangent method. In this method, the onset potential is considered the potential value at which the current begins to deviate noticeably from its baseline. Many studies also employ the threshold method, where the onset potential is defined as the point at which the current exceeds a preset value above the baseline current. Despite these methods being commonly used to determine the onset, none are universally applicable, and their accuracy can depend on the specifics of the experimental system. Furthermore, in many instances, the method used to determine the onset potential is simply not specified [68, 69].

5.1.1 Alloy Formation

In **Paper III**, EQCM analysis was used to determine the onset potential for the formation of the PtHg₄ alloy using a WE consisting of a crystal quartz sensor coated with 200 nm of platinum (25 mm²), in 0.5 M H_2SO_4 , with initial mercury concentration of 10 mg/L. Using the EQCM, the changes in frequency and dissipation were measured for three different harmonics (3, 5, 7)during a linear potential sweep from 0.73 V - 0.18 V vs. SHE. The potential was scanned in a step-wise manner from a high overpotential, where no allow formation would occur, to a sufficiently reductive overpotential that enables the formation of the alloy, while avoiding hydrogen evolution (HER). As the alloy begins to form at the electrode surface, the frequency decreases linearly corresponding to the mass uptake at the crystal surface. Throughout the measurement, the dissipation recorded remains relatively low. However, the slight increase, though minimal, might be indicative of the formation of a soft, viscoelastic layer, possibly representing the formation of elemental mercury at the surface. The subsequent rapid decrease in dissipation suggests a transition of these layers into a more rigid state. This is likely due to the insertion of the elemental mercury into the platinum lattice forming $PtHg_4$ and its diffusion through the growing alloy, as suggested in Section 4.1.

Assuming that the formed alloy demonstrates ideal rigid behavior, the Sauerbrey equation (Eq. 4.2) was used to calculate the mass of the alloy formed on the WE surface from the measured frequency changes. In this case, the mass corresponds to the cumulative current, and the tangent method was used to determine the onset potential, where linear extrapolation was made through the data set ranging from 0.56 V to 0.18 V. As shown in Fig 5.1.b, the onset potential for the alloy formation was found to be 0.63 V vs. SHE.

The formation of the PtHg₄ alloy is a non-spontaneous process. As indicated by the onset potential, the process proceeds very close to the thermodynamic limit, which, based on the Density Functional Theory (DFT) calculations detailed in **Paper III**, is determined to be 0.77 V vs. SHE at a Hg²⁺ concentration of 10 mg/L. For an effective mercury removal, a sufficiently reductive overpotential is essential, while avoiding excessively high potentials that could instigate undesired side reactions such as HER or degrade the electrode material. Within the potential range of 0.63 - 0.18 V vs. SHE, an optimal potential for the alloy formation can be determined. The identification of this optimal potential is crucial to advance the mercury removal technique, ensuring it is not only efficient in removing mercury but also in its energy consumption, a necessity for practical applications.



Figure 5.1: (a) The measured frequency changes and dissipation over time for three different harmonics (f3, f5, and f7), as detected with EQCM during alloy formation. (b) The calculated mass as a function of the applied potential, with the onset potential for the alloy formation determined to be 0.63 V vs. SHE.

5.1.2 Regeneration

In **Paper III**, EQCM analysis was used to determine the onset potential for the regeneration of the platinum WE and the release of the bulk $PtHg_4$ alloy back into the solution. As was done in Section 5.1.1, the changes in frequency and dissipation were measured for three different harmonics (3, 5, 7). The experimental process involved the formation of the alloy at a constant potential of 0.18 V, followed by the application of a constant potential of 0.80 V to remove elemental mercury formed at the surface. Finally, a potential scan from 0.80 V to an higher overpotential of 1.16 V vs. SHE was performed to regenerate the platinum WE and release the bulk alloy, see Fig. 5.2.a. The frequencies measured showed a linear decrease as the alloy was formed, followed by a sharp increase as the elemental mercury at the surface was removed. Subsequently, a linear increase was observed as the bulk alloy was released back into the solution, with the frequency eventually plateauing, indicating the completion of the electrode's regeneration. The changes in dissipation during the formation exhibited a similar trend to that observed in Section 5.1.1. However, the dissipation changes were notably lower when a constant potential was applied to form the alloy, as opposed to a slow potential scan. After the elemental mercury at the surface was removed, the dissipation remained consistent throughout the regeneration process.

As previously done in section 5.1.1, the mass of the alloy was calculated from the measured frequency changes, see fig. 5.2.b. The tangent method, was used to determine the onset potential for the regeneration, where a linear extrapolation was made through the data set ranging from 0.96 V to 1.01 V. As illustrated in Fig. 5.2.c, the onset potential for the regeneration process was identified to be 0.93 V vs. SHE.

Identifying the onset potential for the regeneration of the platinum electrode, and the release of the bulk mercury back into the solution, is particularly important for practical applications. Recovering the mercury from the platinum electrode for secure disposal, and enabling the reuse of the electrodes is a vital aspect of the process. The regeneration should be carried out as quickly and efficiently as possible, which can be facilitated by a higher overpotential. However, applying an excessively high potential to the system could lead to unwanted side-reactions and/or degradation of the electrode material. Further research is needed to identify an ideal "Goldilocks" potential, one that is neither too low nor too high, to achieve optimal regeneration.



Figure 5.2: (a) Measured frequency changes and dissipation over time, for three different harmonics (f3, f5, and f7), as detected with EQCM. (b) The calculated mass as a function of time, and (c) the mass change as a function of the applied potential. The onset potential for the regeneration was determined to be 0.93 V vs. SHE.

5.1.3 Temperature and Concentration Studies

In **Paper II**, batch experiments were conducted using a 2-electrode setup to maintain a clean system, particularly important at low concentrations of mercury where the system is sensitive to any contamination from the RE. These experiments were carried out using a glass WE plated with 100 nm of platinum (15 x 15 mm), in 0.5 M HNO₃ with initial mercury concentrations ranging from 0.25 to 1000 μ g/L, as shown in Fig. 5.3. Notably, over 99% of the mercury was removed from the solution in all cases, or would be, given sufficient removal time. Furthermore, the mercury was reduced to levels far below the WHO's safe limit for drinking water (6 μ g/L) within 30 hours. These concentration studies are essential for evaluating the practicality of the electrochemical removal method for real-world environmental decontamination and sanitation scenarios, where the target mercury concentration is often well below the WHO limit. Owing to the current measurement procedures and the detection limits of the used ICP-MS, experiments with initial mercury concentrations lower than $0.25 \ \mu g/L$ could not be carried out. However, it can be expected that the electrochemical alloy formation will also effectively remove mercury from solutions at substantially lower concentrations.



Figure 5.3: Mercury concentration in solution as a function of time for 2electrode batch measurements. The dashed line corresponds to the WHO guideline for safe drinking water (6 μ g/L).

From the concentration studies, the mercury removal rates were estimated, as shown in Fig. 5.4. It can be observed that the absolute rate of mercury removal is larger at higher initial concentrations, demonstrating a clear concentration dependence. Furthermore, from the general rate expression (Eq. 3.6), both the reaction order (α) and the rate constant (k) can be derived.



Figure 5.4: Natural logarithm of the mercury removal rate as a function of the natural logarithm of the initial mercury concentrations, for 2-electrode batch measurements.

The apparent reaction order in mercury ion concentration was determined to be 0.7. This reaction order of less than one for Hg^{2+} concentration suggests that the alloy formation process does not directly proceed via Hg^{2+} in the solution, contrary to what is indicated by the overall reaction (Eq. 4.1). Instead, it seems more probable that the reaction proceeds via the formation of elemental mercury on the surface, as proposed by Eq. 5.1, and subsequently followed by the alloy formation Eq. 5.2. If indeed the plating of elemental mercury is faster than the alloy formation, elemental mercury will accumulate on the surface, resulting in an apparent reaction order of less than one in Hg^{2+} concentration.

$$Hg^{2+} + 2e^- \to Hg^0 \tag{5.1}$$

$$Pt + 4Hg \rightarrow PtHg_4$$
 (5.2)

In **Paper II**, further investigations were conducted using EQCM analysis at varying temperatures ranging from 20 to 60°C. An Arrhenius plot (Eq. 3.7) was made to determine the apparent activation energy (E_A) of the overall reaction for alloy formation, as depicted in Fig. 5.5.



Figure 5.5: Arrhenius plot for temperatures $20 - 60^{\circ}$ C using EQCM analysis. The error bars represent the standard error of the data for each temperature.

The apparent activation energy for the electrochemical alloy formation was determined to be 0.29 eV. This value being relatively low, i.e. the threshold energy for the reaction to occur is low, is significant for the selective removal of mercury, as the removal process is favored over other potential competing reactions [70]. Notably, the apparent activation energy for alloy formation between platinum and elemental mercury has been reported differently in two separate studies, with values being 0.35 eV [71] and 0.52 eV [48]. The value determined in **Paper II**, 0.29 eV, is relatively close to 0.35 eV, suggesting that the rate-determining step might be the same for both the reaction between elemental mercury and platinum, and the electrochemical alloy formation between mercury ions and platinum.

5.2 Large-scale Mercury Decontamination

Recognizing the need to further develop the electrochemical alloy formation process as a practical method for large-scale mercury decontamination, **Paper I** investigates, for the first time, its potential to remove mercury from concentrated sulfuric acid. For industrial application, the method should be scalable to treat hundreds or even thousands of tonnes of concentrated sulfuric acid. Furthermore, it should effectively lower the mercury concentration in the acid to levels that are considered technical quality grade (0.30 mg/kg) or high-purity grade (0.08 mg/kg) [72], within specified time frames.

For large-scale decontamination, it is essential for the WE to possess a high surface area support onto which platinum can be deposited, in order to increase the rate of mercury removal, while reducing the amount of platinum needed. Identifying appropriate materials for the WE posed a substantial engineering challenge, primarily due to sulfuric acid's highly corrosive nature. Given their high surface area foam materials emerged as suitable candidates for the WE. Specifically, Stainless Steel 316L (SS316L) foam, commonly used in the industrial storage and handling of concentrated sulfuric acid, was identified as a potential support material [59, 73]. In this study, five different SS316L foams (25 x 15 mm), listed in Table 5.1, each with varying porosity and coated with 2 μ m of platinum from two reputable manufacturers, were evaluated.

Table 5.1: Five different SS316L foams ($15 \times 25 \text{ mm}$), with different porosity, each coated with 2 µm of platinum. The foams originate from two different manufacturers, referred to as A and B. Additional platinum powder coating was applied to the fifth and final foam, as denoted by (*).

WE foam	\mathbf{PPI}	Manufacturer
F1	45	А
F2	80	В
F3	30	А
F4	60	А
F5	45^{*}	В

Preliminary batch experiments were conducted at lab-scale using the different SS316L foams as support material for the WE. Mercury was successfully removed from samples of authentic sulfuric acid from a zinc smelter, as shown in Fig. 5.6.a. In all cases (F1-F5), mercury was reduced below the technical limit within 80 hours, with F1 outperforming the rest by achieving high purity quality of the acid within 100 hours. However, all SS316L foams exhibited instability in the concentrated acid during mercury removal, as evidenced by iron leaching from the electrodes, see Fig.5.6.b. For all five foams, the iron concentration in the acid exceeded the high purity limit (15 mg/kg). Moreover, for foams F1, F4, and F5, the iron concentration surpassed the technical quality limit (30 mg/kg), with F4 exhibiting the most significant instability. This instability was unexpected and suggests that the foams might not possess the stated high-quality SS316L structure, possibly due to defects from the fabrication or the high-temperature platinum plating process. The observed instability does not appear to correlate with the manufacturer or the pore size of the foam.



Figure 5.6: (a) Mercury removal from concentrated sulfuric acid at lab-scale (50 mL), using five different SS316L foams (15 x 25 mm). (b) The stability of all five foam during the mercury removal in terms of iron concentration in the acid. The curve fittings serve as a visual aid without representing the best fit of the data spread.

Despite the SS316L foam (F1) lacking stability in the concentrated acid at lab-scale, the mercury removal was still excellent, and thus, the foam was used to further evaluate the scalability of the removal method in a 20 L scale-up. The WE was also up-scaled from one SS316L foam (15 x 25 mm) to two SS316L WEs (200 x 150 mm). The results, which demonstrate the decrease in mercury concentration and the corresponding increase in iron concentration, during the mercury removal process, are shown in Fig. 5.7.



Figure 5.7: (a) Mercury removal from concentrated sulfuric acid using platinum SS316L foam WE, for both mercury removal at lab-scale (50 mL) and in a scaleup (20 L). (b) The mercury removal in the pilot reactor and the corresponding increase in iron concentration. The curve fittings serve only as visual aids without representing the best fit of the data spread.

Mercury was successfully removed from concentrated sulfuric in the scaleup, where over 98% of the mercury content was removed from the acid with initial concentration of approximately 0.8 mg/kg, within 160 hours. Technical quality of the acid was reached after around 20 hours, and high purity quality within 80 hours. The removal trends at the lab-scale and in the scale-up align after 60 hours, with high-purity quality achieved within 80 hours in both cases. Differences in removal behavior can be attributed to factors such as cathode electrode area vs. electrolyte volume, temperature fluctuations, number and size of anode electrodes, and reactor geometry. Despite these differences, the removal process largely follows the same trend and achieves mercury concentrations well below technical and high purity quality. Iron contamination, which initially exceeded high-purity limits, increased above technical quality after 160 hours. This confirmed scalability indicates that the method has great potential for industrial-scale applications, treating thousands of tonnes of concentrated sulfuric acid, which has never been demonstrated before. However, moving forward, an alternative support material is needed for the WE, as SS316L was discovered to have poor corrosion resistance to the concentrated acid and thus unsuitable for future applications.

To prevent potential iron contamination of the acid, a metal-free candidate, a reticulated vitreous carbon (RVC) was used as an alternative to SS316L [60]. The RVC foam (15 x 40 mm), plated with 100 nm of platinum, demonstrated excellent stability in the concentrated sulfuric acid at lab-scale, where around 90% of the mercury content was removed from the acid within 120 hours, see Fig. 5.8. Technical quality of the acid was reached within 20 hours, however the mercury concentration appears to unexpectedly plateau and not decrease below 0.10 mg/kg, just above the high-purity quality. This may be due to the nature of platinum deposition on porous substrates, which can result in uneven film thickness. As the removal process continues, areas with thinner platinum layers become saturated, leading to less available platinum surface area for mercury absorption and consequently, lower removal rates. Furthermore, the RVC foam was only plated with platinum on one side, limiting the total available surface area compared to the SS316L foam, which was coated on all sites. With improved coating techniques, it is likely that the RVC foam could achieve mercury concentrations below high-purity targets.



Figure 5.8: Mercury removal from concentrated sulfuric acid at lab-scale (50 mL) using RVC foam (15×40 mm) WE. The curve fittings serve as a visual aid without representing the best fit of the data spread.

Chapter 6 Conclusion and Outlook

The research conducted in this thesis provides substantial insights into electrochemical alloy formation as a technique for mercury removal. A series of studies demonstrated effective mercury removal from solutions with initial mercury concentrations as low as $0.25 \ \mu g/L$. This proves the potential of the method to reduce mercury to extremely low levels in a relatively short amount of time, highlighting its applicability for decontamination of natural waters. In addition, the alloy formation process was found to have an apparent reaction order below one in Hg^{2+} concentration, which is an important kinetic parameter for understanding the alloy formation mechanism and its performance at specific mercury concentrations. The temperature dependence and the determined apparent activation energy of 0.29 eV are also crucial considerations for practical applications, as temperatures can vary significantly across different industrial streams. Furthermore, by using electrochemical alloy formation on platinum, mercury was removed from concentrated sulfuric acid for the first time. Over 98% of mercury was removed from the acid with initial concentration of approximately 0.8 mg/kg, resulting in high quality acid with mercury content below key industrial limits of technical quality (0.30 mg/kg) and high purity (0.08 mg/kg). This method proved effective at both 50 mL lab-scale and in a 20 L scale-up, demonstrating its excellent scalability and potential for large-scale industrial applications.

While this mercury removal technique holds potential, it also comes with its own set of challenges and limitations. One general challenge, not limited to the present method, is that the analysis of mercury in solution is complex, even when sophisticated analytical techniques such as ICP-MS, which is widely used in clinical laboratories, are employed. Mercury is well-known for its memory effect, which can lead to an overestimation of mercury levels. Rigorous sample preparation is therefore essential, including ensuring mercury stability in the samples being evaluated and adjusting for correct dilution within the detection limits of the instrument. Furthermore, it is important to tailor the sample matrices to accurately reflect the environment being studied to ensure dependable and accurate results. A significant challenge lies in identifying the ideal support material for the platinum electrode in larger-scale applications. While the SS316L foam demonstrated effectiveness in mercury removal, its poor corrosion resistance makes it unsuitable. In contrast, the carbon foam (RVC) displayed both efficient mercury removal and notable corrosion resistance, marking it as a promising candidate for the platinum electrode support material. However, further investigations are necessary to fully evaluate the materials stability, longevity, and effectiveness over numerous cycles of mercury removal and regeneration. This is particularly important as the electrodes must endure repeated mercury extraction processes without performance degradation or contamination of the sulfuric acid, to be considered viable for commercial applications. To develop this removal technique into a viable decontamination solution, every stage of the process, including the regeneration, must be thoroughly designed and optimized. In particular, future research focusing in more depth on the regeneration process is essential for establishing a comprehensive end-to-end mercury removal system.

In conclusion, the electrochemical alloy formation has shown to be a promising method for mercury removal. With further research, optimization, and technological advancements, it has the potential to be an effective solution for mercury decontamination of aqueous systems, with real-life applications in environmental remediation, industrial waste treatment, large-scale decontamination of sulfuric acid in the mining industry, and many more.

Acknowledgment

This project is financially supported by the Swedish Research Council for Sustainable Development (Formas). Additional support was obtained from the Strategic Innovation Program Swedish Mining Innovation, funded by Vinnova and the Swedish Energy Agency.

First and foremost, I would like to thank my supervisor, Björn Wickman, for his support, patience, and insightful feedback, which guided me throughout this project. I'm also grateful to my co-supervisor, Teodora Retegan Vollmer, for her support and expertise. My sincere thanks to my examiner, Henrik Grönbeck, for his constructive critique and input.

To all the members of the electrochemistry group and the team at Atium, thank you for all the engaging discussions, continuous encouragements and friendship. Finally, I want to express my gratitude to my family and friends, both in Sweden and back home in Iceland.

Bibliography

- B. Ericson, J. Caravanos, K. Chatham-Stephens, P. Landrigan and R. Fuller, "Approaches to systematic assessment of environmental exposures posed at hazardous waste sites in the developing world: The toxic sites identification program," *Environmental Monitoring and Assessment*, vol. 185, pp. 1755–1766, 2 2012. DOI: 10.1007/s10661-012-2665-2.
- [2] "World's worst pollution problems report," Blacksmith Institute and Green Cross Switzerland, 2016.
- [3] G. Filippelli, S. Anenberg, M. Taylor, A. Geen and H. Khreis, "New approaches to identifying and reducing the global burden of disease from pollution," *GeoHealth*, vol. 4, 4 Apr. 2020. DOI: 10.1029/2018GH000167.
- [4] "World's worst pollution problems," Pure Earth and Green Cross Switzerland, 2015.
- [5] "Guidelines for drinking-water quality, 4th edition, incorporating the 1st addendum," WHO, 2017.
- [6] "Minamata convention on mercury: Annotated bibliography of WHO information," World Health Organization, Publications, 2021, iv, 39 p.
- [7] N. Pirrone et al., "Global mercury emissions to the atmosphere from natural and anthropogenic sources," Mercury Fate and Transport in the Global Atmosphere, pp. 1–47, 2009. DOI: 10.1007/978-0-387-93958-2_1.
- [8] D. G. Streets *et al.*, "Total mercury released to the environment by human activities," *Environmental Science Technology*, vol. 51, pp. 5969–5977, 11 2017. DOI: 10.1021/acs.est.7b00451.
- [9] D. O'Connor *et al.*, "Mercury speciation, transformation, and transportation in soils, atmospheric flux, and implications for risk management: A critical review," *Environment International*, vol. 126, pp. 747–761, 2019. DOI: 10.1016/j.envint.2019.03.019.
- [10] N. E. Selin, "Global biogeochemical cycling of mercury: A review," Annual Review of Environment and Resources, vol. 34, pp. 43–63, 1 2009. DOI: 10.1146/annurev.environ.051308.084314.

- P. A. Ariya *et al.*, "Mercury physicochemical and biogeochemical transformation in the atmosphere and at atmospheric interfaces: A review and future directions," *Chemical Reviews*, vol. 115, pp. 3760–3802, 10 2015. DOI: 10.1021/cr500667e.
- [12] A Sharma, A Sharma and R. K. Arya, "Removal of mercury(ii) from aqueous solution: A review of recent work," *Separation Science and Technology (Philadelphia)*, vol. 50, pp. 1310–1320, 9 2015, Cited By :2 Export Date: 26 February 2016. DOI: 10.1080/01496395.2014.968261.
- [13] K. Hua, X. Xu, Z. Luo, D. Fang, R. Bao and J. Yi, "Effective removal of mercury ions in aqueous solutions: A review," *Current Nanoscience*, vol. 16, pp. 363–375, 3 2020. DOI: 10.2174/1573413715666190112110659.
- G. Crini and E. Lichtfouse, "Advantages and disadvantages of techniques used for wastewater treatment," *Environmental Chemistry Letters*, 2018. DOI: 10.1007/s10311-018-0785-9.
- [15] C. Tunsu and B. Wickman, "Effective removal of mercury from aqueous streams via electrochemical alloy formation on platinum," *Nature Communications*, vol. 9, 2018. DOI: 10.1038/s41467-018-07300-z.
- [16] M. K. O. Bengtsson, C. Tunsu and B. Wickman, "Decontamination of mercury-containing aqueous streams by electrochemical alloy formation on copper," *Industrial Engineering Chemistry Research*, vol. 58, pp. 9166– 9172, 21 2019. DOI: 10.1021/acs.iecr.9b01513.
- W. Fitzgerald and C. Lamborg, "9.04 geochemistry of mercury in the environment," in *Treatise on Geochemistry*, H. D. Holland and K. K. Turekian, Eds., Oxford: Pergamon, 2007, pp. 1–47. DOI: 10.1016/B0-08-043751-6/09048-4.
- [18] U. N. E. Programme, "Global mercury assessment 2013: Sources, emissions, releases and environmental transport," 2013.
- [19] C. T. Driscoll, R. P. Mason, H. M. Chan, D. J. Jacob and N. Pirrone, "Mercury as a global pollutant: Sources, pathways, and effects," *Envir*onmental Science Technology, vol. 47, pp. 4967–4983, 10 2013. DOI: 10.1021/es305071v.
- [20] A. B. Mukherjee, R Zevenhoven, J Brodersen, L. D. Hylander and P Bhattacharya, "Mercury in waste in the european union: Sources, disposal methods and risks," *Resources Conservation and Recycling*, vol. 42, pp. 155–182, 2 2004. DOI: 10.1016/j.resconrec.2004.02.009.
- [21] L. D. Hylander and R. B. Herbert, "Global emission and production of mercury during the pyrometallurgical extraction of nonferrous sulfide ores," *Environmental Science Technology*, vol. 42, pp. 5971–5977, 16 Aug. 2008. DOI: 10.1021/es800495g.
- [22] "Arctic pollution," Arctic Monitoring and Assessment Programme, 2011.
- [23] W. Shotyk, "Arctic plants take up mercury vapour," Nature, vol. 547, pp. 167–168, 7662 Jul. 2017. DOI: 10.1038/547167a.

- [24] K. G. Pavithra, P. SundarRajan, P. S. Kumar and G. Rangasamy, "Mercury sources, contaminations, mercury cycle, detection and treatment techniques: A review," *Chemosphere*, vol. 312, p. 137 314, Jan. 2023. DOI: 10.1016/j.chemosphere.2022.137314.
- [25] N. Selin, D. Jacob, R. Yantosca, S. Strode, L. Jaeglé and E. Sunderland, "Global 3-d land-ocean-atmosphere model for mercury: Presentday versus preindustrial cycles and anthropogenic enrichment factors for deposition," *Global Biogeochemical Cycles*, vol. 22, Sep. 2008. DOI: 10.1029/2007GB003040.
- S. M. Ullrich, T. W. Tanton and S. A. Abdrashitova, "Mercury in the aquatic environment: A review of factors affecting methylation," *Critical Reviews in Environmental Science and Technology*, vol. 31, pp. 241–293, 3 2001. DOI: 10.1080/20016491089226.
- [27] U. environment, "Global mercury assessment," Chemicals and Health Branch International Environment House, 2018.
- [28] N. E. Selin, "Mercury rising: Is global action needed to protect human health and the environment?" *Environment: Science and Policy for Sustainable Development*, vol. 47, no. 1, pp. 22–35, 2005. DOI: 10.3200/ ENVT.47.1.22-37.
- [29] M. F. Wolfe, S. Schwarzbach and R. A. Sulaiman, "Effects of mercury on wildlife: A comprehensive review," *Environmental Toxicology and Chemistry*, vol. 17, pp. 146–160, 2 Feb. 1998. DOI: 10.1002/etc.5620170203.
- [30] A. Scheuhammer, "The chronic toxicity of aluminium, cadmium, mercury, and lead in birds: A review," *Environmental Pollution*, vol. 46, pp. 263– 295, 4 1987, ISSN: 02697491. DOI: 10.1016/0269-7491(87)90173-4.
- [31] J.-D. Park and W. Zheng, "Human exposure and health effects of inorganic and elemental mercury," *Journal of Preventive Medicine Public Health*, vol. 45, pp. 344–352, 6 Nov. 2012. DOI: 10.3961/jpmph.2012.45.6.344.
- [32] B. F. Azevedo *et al.*, "Toxic effects of mercury on the cardiovascular and central nervous systems," *Journal of Biomedicine and Biotechnology*, vol. 2012, pp. 1–11, 2012. DOI: 10.1155/2012/949048.
- [33] J. C. Clifton, "Mercury exposure and public health," *Pediatric Clinics of North America*, vol. 54, pp. 237-+, 2 2007. DOI: 10.1016/j.pcl.2007.02.005.
- [34] Toxicological Effects of Methylmercury. National Academies Press, Sep. 2000, ISBN: 978-0-309-07140-6. DOI: 10.17226/9899.
- [35] M. Harada, "Minamata disease: Methylmercury poisoning in japan caused by environmental pollution," *Critical Reviews in Toxicology*, vol. 25, pp. 1– 24, 1 Jan. 1995. DOI: 10.3109/10408449509089885.
- [36] B. Institute and G. C. Switzerland, "World's worst pollution problems report 2010," 2010.

- [37] "Council directive 98/83/ec of 3 november 1998 on the quality of water intended for human consumption," Official Journal of the European Communities, vol. L 330/32, 1998.
- [38] F. L. Fu and Q Wang, "Removal of heavy metal ions from wastewaters: A review," Journal of Environmental Management, vol. 92, pp. 407-418, 3 2011. DOI: 10.1016/j.jenvman.2010.11.011.
- [39] L. Wang et al., "Remediation of mercury contaminated soil, water, and air: A review of emerging materials and innovative technologies," *Environment International*, vol. 134, p. 105 281, 2020. DOI: 10.1016/j.envint.2019. 105281.
- [40] G. Crini and E. Lichtfouse, "Advantages and disadvantages of techniques used for wastewater treatment," *Environmental Chemistry Letters*, vol. 17, pp. 145–155, 1 Mar. 2019. DOI: 10.1007/s10311-018-0785-9.
- [41] J. Patterson, E. Barth and L. Stein, "Aqueous mercury treatment," Jul. 1997.
- [42] J. G. Yu et al., "Removal of mercury by adsorption: A review," Environmental Science and Pollution Research, vol. 23, pp. 5056–5076, 6 2016.
 DOI: 10.1007/s11356-015-5880-x.
- [43] M. Taseidifar, F. Makavipour, R. M. Pashley and A. M. Rahman, "Removal of heavy metal ions from water using ion flotation," *Environmental Technology Innovation*, vol. 8, pp. 182–190, 2017. DOI: https: //doi.org/10.1016/j.eti.2017.07.002.
- [44] J. A. Ritter and J. P. Bibler, "Removal of mercury from waste-water large-scale performance of an ion exchange process," *Water Science and Technology*, vol. 25, pp. 165–172, 3 1992.
- [45] C. H. Hamann, A. Hamnett and W. Vielstich, *Electrochemistry*, 2nd. WILEY-VCH, 2007, p. 550.
- [46] A. M. Hartley, A. G. Hiebert and J. A. Cox, "Preparation and properties of a platinum-based mercury-film electrode," *Journal of Electroanalytical Chemistry and Interfacial Electrochemistry*, vol. 17, pp. 81–86, 1 1968. DOI: 10.1016/S0022-0728(68)80032-4.
- [47] M. Z. Hassan, D. F. Untereker and S. Bruckenstein, "Ring-disk study of thin mercury films on platinum," *Journal of Electroanalytical Chemistry* and Interfacial Electrochemistry, vol. 42, pp. 161–181, 2 1973. DOI: 10. 1016/S0022-0728(73)80390-0.
- S. K. Lahiri and D. Gupta, "A kinetic study of platinum-mercury contact reaction," *Journal of Applied Physics*, vol. 51, pp. 5555–5560, 10 1980.
 DOI: 10.1063/1.327440.
- [49] G. C., "The Hg-Pt (mercury-platinum) system," Bulletin of Alloy Phase Diagrams, vol. 11, pp. 26–32, 1 1990. DOI: 10.1007/BF02841581.
- [50] F. L. Fertonani, A. V. Benedetti and M Ionashiro, "Contribution to the study of the reaction of mercury with platinum and a platinumiridium alloy," *Thermochimica Acta*, vol. 265, pp. 151–161, 1995. DOI: 10.1016/0040-6031(95)02417-Z.

- [51] G. R. Souza, I. A. Pastre, A. V. Benedetti, C. A. Ribeiro and F. L. Fertonani, "Solid state reactions in the platinum-mercury system thermogravimetry and differential scanning calorimetry," *Journal of Thermal Analysis* and Calorimetry, vol. 88, pp. 127–132, 2007. DOI: 10.1007/s10973-006-8037-9.
- [52] H. Okamoto and T. B. Massalski, "The Au-Hg (gold-mercury) system," Bulletin of Alloy Phase Diagrams, vol. 10, pp. 50–58, 1 Feb. 1989. DOI: 10.1007/BF02882176.
- [53] Q. ul-ain khan et al., "Synthesis and mechanical properties of dental amalgam," Materials Today: Proceedings, vol. 47, S33-S37, 2021. DOI: 10.1016/j.matpr.2020.04.672.
- [54] U. G. Bengtsson and L. D. Hylander, "Increased mercury emissions from modern dental amalgams," *BioMetals*, vol. 30, pp. 277–283, 2 Apr. 2017. DOI: 10.1007/s10534-017-0004-3.
- [55] D. A. Buttry and M. D. Ward, "Measurement of interfacial processes at electrode surfaces with the electrochemical quartz crystal microbalance," *Chemical Reviews*, vol. 92, pp. 1355–1379, 6 1992. DOI: 10.1021/ cr00014a006.
- [56] S. Nilsson, F. Björefors and N. D. Robinson, "Electrochemical quartz crystal microbalance study of polyelectrolyte film growth under anodic conditions," *Applied Surface Science*, vol. 280, pp. 783–790, Sep. 2013. DOI: 10.1016/j.apsusc.2013.05.062.
- [57] E. J. Calvo and R. Etchenique, Kinetic applications of the electrochemical quartz crystal microbalance (eqcm), 1999. DOI: 10.1016/S0069-8040(99) 80017-X.
- [58] G. Sauerbrey, "Verwendung von schwingquarzen zur wägung dünner schichten und zur mikrowgung," Zeitschrift für Physik, vol. 155, pp. 206– 222, 2 Apr. 1959. DOI: 10.1007/BF01337937.
- [59] W. Gunnar, "Metal foams for electrodes," Fraunhofer Institute for Manufacturing Technology and Advanced Materials IFAM Branch Lab, 2022.
- [60] E. Aerospace, Carbon foam. rvc foam. open cell foam, reticulated foam. [Online]. Available: https://ergaerospace.com/.
- [61] S. Wilschefski and M. Baxter, "Inductively coupled plasma mass spectrometry: Introduction to analytical aspects," *Clinical Biochemist Reviews*, vol. 40, pp. 115–133, 3 Aug. 2019. DOI: 10.33176/AACB-19-00024.
- [62] C. F. Harrington, S. A. Merson and T. M. D. Silva, "Method to reduce the memory effect of mercury in the analysis of fish tissue using inductively coupled plasma mass spectrometry," *Analytica Chimica Acta*, vol. 505, pp. 247–254, 2 Mar. 2004. DOI: 10.1016/j.aca.2003.10.046.
- [63] B. Mei, W. Fong, S. Siu, J. Sai, K. Tee and S. Tam, "Determination of mercury in whole blood and urine by inductively coupled plasma mass spectrometry," 2007.

- [64] J. L. Rodrigues *et al.*, "Determination of total and inorganic mercury in whole blood by cold vapor inductively coupled plasma mass spectrometry (cv icp-ms) with alkaline sample preparation," *Journal of Analytical Atomic Spectrometry*, vol. 24, pp. 1414–1420, 10 2009. DOI: 10.1039/ b910144f.
- [65] R. S. Pappas, "Sample preparation problem solving for inductively coupled plasma-mass spectrometry with liquid introduction systems i. solubility, chelation, and memory effects."
- [66] Y. Li *et al.*, "Elimination efficiency of different reagents for the memory effect of mercury using icp-ms," *Journal of Analytical Atomic Spectrometry*, vol. 21, pp. 94–96, 1 2006. DOI: 10.1039/b511367a.
- [67] M. Korvela, M. Andersson and J. Pettersson, "Internal standards in inductively coupled plasma mass spectrometry using kinetic energy discrimination and dynamic reaction cells," *Journal of Analytical Atomic Spectrometry*, vol. 33, pp. 1770–1776, 10 2018. DOI: 10.1039/C8JA00171E.
- [68] C. Batchelor-McAuley, "Defining the onset potential," Current Opinion in Electrochemistry, vol. 37, p. 101176, Feb. 2023. DOI: 10.1016/j. coelec.2022.101176.
- [69] G. de Falco, M. Florent, A. D. Rosa and T. J. Bandosz, "Proposing an unbiased oxygen reduction reaction onset potential determination by using a savitzky-golay differentiation procedure," *Journal of Colloid and Interface Science*, vol. 586, pp. 597–600, Mar. 2021. DOI: 10.1016/j. jcis.2020.10.127.
- [70] A. L. Petrou and M Economou-Eliopoulos, "The activation energy values estimated by the arrhenius equation as a controlling factor of platinumgroup mineral formation," *Geochimica Et Cosmochimica Acta*, vol. 73, pp. 1625–1636, 6 2009. DOI: 10.1016/j.gca.2008.12.009.
- [71] M Barlow and P. J. Planting, "Heterometallic phenomena in platinummercury system," *Zeitschrift Fur Metallkunde*, vol. 60, 292–, 4 1969, Barlow, m planting, pj.
- [72] "Sulphur products," Boliden Metals, 2018.
- [73] F. Roger, "The performance of stainless steels in concentrated sulphuric acid," Stainless Steel World, 2009.